Author's response to reviews

Title: Topical Clobetasol for the Treatment of Toxic Epidermal Necrolysis: Study Protocol for a Randomized Controlled Trial

Authors:

Reason Wilken (reason.wilken@gmail.com)
Chin-Shang Li (cssli@ucdavis.edu)
Victoria R Sharon (Victoria.sharon@ucdmc.ucdavis.edu)
Kyoungmi Kim (kmkim@ucdavis.edu)
Falin B Patel (Falin.patel@gmail.com)
Forum Patel (patelforumb@gmail.com)
Emanual Maverakis (emaverakis@ucdavis.edu)

Version: 7 Date: 17 July 2015

Author's response to reviews: see over
Dear Trials,

Thank you very much for your review of our proposed clinical trial protocol “Topical clobetasol for the treatment of toxic epidermal necrolysis: study protocol for a randomized placebo-controlled trial” (MS # 1149667164138970) that was submitted in January 2015 and subsequently re-submitted with revisions in April 2015 and June 2015. We have received the reviewer comments from the journal regarding the revised manuscript and our point by point responses are provided below:

1. **Title:** Altered the title in the submission system to conform to journal format and match the title on the manuscript “Topical clobetasol for the treatment of toxic epidermal necrolysis: study protocol for a randomized controlled trial”.

2. **Abstract:** Replaced the abstract in the submission system to be identical to the abstract in the manuscript. In addition, the trial registration number and date was uploaded into the submission system below the abstract in the manuscript.

3. **Keywords:** Additional section containing keywords was added to the manuscript following the trial registration information.

Thank you again for your review of our proposed trial protocol. Attached we have submitted the revised manuscript and corrected the above items in the submission system. We look forward to hearing any further comments and suggested changes for manuscript revision in order to make it suitable for publication in your esteemed journal.

Sincerely,

Emanual Maverakis, M.D.
University of California, Davis School of Medicine
Department of Dermatology- Associate Professor
3301 C Street, Suite 1400
Sacramento, CA 95816
Telephone (916) 843-7336
Fax (916) 843-9444
emaverakis@ucdavis.edu